Author's response to reviews


Authors:

Rachel Eyre (Rachel.Eyre@ncl.ac.uk)
Richard G Feltbower (R.G.Feltbower@ncl.ac.uk)
Peter W James (P.W.James@ncl.ac.uk)
Karen Blakey (Karen.Blakey@ncl.ac.uk)
Emmanuel Mubwandarikwa (e.mubwandarikwa@tiscali.co.uk)
David Forman (formand@iarc.fr)
Patricia A McKinney (P.A.McKinney@leeds.ac.uk)
Mark S Pearce (M.S.Pearce@ncl.ac.uk)
Richard JQ McNally (Richard.McNally@ncl.ac.uk)

Version: 10 Date: 5 July 2010

Author's response to reviews:

I have made the following minor revisions:

1. Reference 25 - I have aligned the final line of the reference.
2. Reference 26 - Missing author initial - I have changed "Renard" to "Renard H".
3. I have closed the gap between References 31 & 32.